Author’s response to reviews

Title: Treatment of thoraco-lumbar curves in adolescent females affected by idiopathic scoliosis with a progressive action short brace (PASB): assessment of results according to the SRS committee on bracing and nonoperative management standardization criteria.

Authors:

Angelo G. Aulisa (agaulisa@tiscali.it)
Vincenzo Guzzanti (vguzzanti@yahoo.it)
Marco Galli (dottorgalli@yahoo.it)
Carlo Perisano (carloperisano@hotmail.it)
Francesco Falciglia (falciglia@opbg.net)
Lorenzo Aulisa (aulisa@libero.it)

Version: 6 Date: 11 September 2009

Author's response to reviews:

Dear Prof. TB Grivas
I have made all the formatting changes requested.
Kind regards
Dr A. G. Aulisa